Author’s response to reviews

**Title:** Formative Research for the Design of a Scalable Mobile Health Program on Water, Sanitation, and Hygiene: CHoBI7 Mobile Health Program

**Authors:**

Christine George (cgeorg19@jhu.edu)
Fatema Zohura (fzohura@icddrb.org)
Alana Teman (alana.teman@gmail.com)
Elizabeth Thomas (liz.thomas@jhu.edu)
Tasdik Hasan (tasdikhdip@yahoo.com)
Sohel Rana (mrana5@jhu.edu)
Tahmina Parvin (tparvin@icddrb.org)
David Sack (dsack1@jhu.edu)
Md. Sazzadul Bhuyian (sazzadul.islam@icddrb.org)
Alain Labrique (alabrique@gmail.com)
Jahed Masud (jahed@icddrb.org)
Kelsey Zeller (zeller.kelsey@gmail.com)
Farzana Begum (farzanab@icddrb.org)
Md. Abul Khan (cbhc@ld.dghs.gov.bd)
Sanya Tahmina (sanya.tahmina@gmail.com)
Farzana Munmun (munmun854@gmail.com)
Shirajum Monira (smonira@icddrb.org)
Munirul Alam (munirul@icddrb.org)

**Version:** 2  **Date:** 05 Jun 2019

Author’s response to reviews:
Dear Editorial Board,

We have addressed you below comments.

1. Please declare if the other authors have no competing interest in the "Competing Interest" section.
   
   RESPONSE: Updated

2. Please remove figure 2 in the figure legends section.
   
   RESPONSE: Updated

Regards,

Christine Marie

---

Christine Marie George, PhD
Associate Professor
Department of International Health
Program in Global Disease Epidemiology and Control
Johns Hopkins Bloomberg School of Public Health
615 N. Wolfe Street, Room E5535
Baltimore, Maryland 21205-2103
Telephone: (410) 955-2485